Author’s response to reviews

Title: Dynamic prediction of long-term survival in patients with primary gastric diffuse large Bcell lymphoma: A SEER population-based study

Authors:

Ju-Li Lin (632366024@qq.com)
Jian-Xian Lin (linjian379@163.com)
Ping Li (24627878@qq.com)
Jian-Wei Xie (364531721@qq.com)
Jia-bin Wang (847044493@qq.com)
Jun Lu (78379048@qq.com)
Qi-Yue Chen (690934662@qq.com)
Long-long Cao (1291821982@qq.com)
Chao-Hui Zheng (wwkzch@163.com)
Chang-ming Huang (hcmlr2002@163.com)

Version: 5 Date: 23 Jul 2019

Author’s response to reviews:

Editor Comments:

1. Thank you for clarifying that consent to participate was not required as "Because patient data in the SEER database were de-identified, signed informed consent was waived for the present study". Please include a statement to this effect in your 'Ethics Approval and Consent to Participate' section.

Response: We have revised the 'Ethics Approval and Consent to Participate' section as you request.

2. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional
files) should remain uploaded as separate files. Please ensure that all figures, tables and additional/supplementary files are cited within the text.

Response: We have revised the manuscript as you request.